\begin{table}%T7
\caption{Iron abundances of two Hyades giants.}
\label{hyades}
%\centerline
{
\begin{tabular}{rlllllllllll}
\hline\hline
\multicolumn{6}{c}{HD 27371}&\multicolumn{6}{c}{HD 27697}\\
\hline
Ref. & \teff  & log g & $\xi$ & \hfill [FeI/H] \hfill & [FeII/H] & 
Ref. & \teff  & log g & $\xi$ & \hfill [FeI/H] \hfill & [FeII/H] \\
\hline
1 & 4880 & 2.50 & 1.3 & $+0.21\pm 0.13$ & +0.15 & 1 & 4880 & 2.50 & 1.4 &
$+0.23\pm 0.13$ & +0.05\\ 
2 & 5143  & 3.02 &    &  +0.07          &       & 2  & 5143 & 2.98 &  & +0.08\\
3 & 5040  & 2.58 &    &  +0.17          &       & 3  & 5040 & 2.44 &  & +0.41\\
4 & 5663  & 2.0  &    &  $-0.26$          &       & 4  &                       \\
5 & 4271  & 3.0  &    &  +0.34          &       & 5  & 4235  & 3.0  & &  +0.30\\
6 & 4990  & 2.80 &    &  +0.20          &       & 6       \\
7 & 4800  & 2.60 &    &  +0.09          &       & 7  \\
8 & 4930  & 2.90 &    &  $-0.02$          &       & 8  & 4940  & 2.85 & &  +0.00\\
9 & 4900  & 2.60 &    &  +0.13          &       & 9  & 4875  & 2.40 & &  +0.06\\
10& 4965  & 2.65 &    &  $+0.15\pm 0.03$ &       \\
\hline
\end{tabular}
}
\par
\medskip
 (1) This paper, (2) \citet{lambert}, (3) \citet{gratto82},
(4) \citet{Koma80}, (5) \citet{Koma85}, (6) \citet{gratto86},
(7) \citet{Fernan}, (8) \cite{mc90}, (9) \citet{Luck},
(10) \citet{Smith}.
\end{table}